Author’s response to reviews

Title: Accuracy of anthropometric measurements by general practitioners in overweight and obese patients

Authors:

Paul Sebo (paulsebo@hotmail.com)

Dagmar Haller (dagmar.haller-ester@hcuge.ch)

François Herrmann (francois.herrmann@hcuge.ch)

Version: 1 Date: 13 Oct 2016

Author’s response to reviews:

We modified "Ethical approval and consent to participate" section according to editor comments:

Ethical approval and consent to participate: The research protocol was approved by the local research ethics committee (Commission d’éthique pour la recherche clinique en ambulatoire). To take part in the study, both GPs and volunteers had to read the information sheet and sign consent forms.